Author's response to reviews

Title: Predictors of short-term outcome of corticosteroid injections in rotator cuff disease, a prospective study

Authors:

Ole M Ekeberg (o.m.ekeberg@medisin.uio.no)
Erik Bautz-Holter (erik.bautz-holter@medisin.uio.no)
Niels G Juel (NielsGunnar.Juel@ulleval.no)
Kaia B Engebretsen (k.b.engebretsen@medisin.uio.no)
Synnøve Kvalheim (kvsy@ulleval.no)
Jens I Brox (jens.ivar.brox@oslo-universitetssykehus.no)

Version: 2 Date: 12 April 2010

Author's response to reviews:

BMC musculoskeletal disorders.

Editor

Date: 12.04.10

Submission of revised manuscript.

Enclosed is a revised prospective research article entitled “Predictors of short-term outcome of corticosteroid injections in rotator cuff disease – a prospective study”. I was asked to report the trial registration number. The study is a secondary analysis of a randomized trial that is registered in the clinical trials register (Clinical trials NCT00640575). I have now included this trial number in the abstract. Please take contact if I need to obtain a unique trial number for this paper.

This manuscript represents unpublished material and does not contain results previously published in text, illustrations, or tables. This manuscript is not under consideration for publication elsewhere, and further, has not been posted to the Internet for public access.

Yours sincerely,

Ole M. Ekeberg, MD, Research fellow